Author's response to reviews

Title: NKTR-102 versus irinotecan as an approach for the treatment of brain metastasis of breast cancer

Authors:

Chris E Adkins (ciadkins@hsc.wvu.edu)
Mohamed I Nounou (nounou@alexu.edu.eg)
Tanvirul Hye (tanvirul.hye@ttuhsc.edu)
Afroz S Mohammad (asmohammad@mix.wvu.edu)
Tori Terrell-Hall (tbterrell@mix.wvu.edu)
Neel K Mohan (nmohan@nektar.com)
Michael A Eldon (meldon@nektar.com)
Ute Hoch (UHoch@nektar.com)
Paul R Lockman (prlockman@hsc.wvu.edu)

Version: 2
Date: 19 March 2015

Author's response to reviews:

Response to requested changes to document prior to editorial and/or peer review:

1) Line numbering: we have now added line numbering.

2) Ethics: This work did not involve/include human subject. This study involved mice only; the ethical treatment and names of the bodies which granted approval for this work are described in the methods section.